

Great binding affinity of SARS-CoV-2 proteins to resin composite monomers and polymerised chains

Sette-de-Souza PH *et al.* SARS-CoV-2 and Resin Composite interactions

Pedro Henrique Sette-de-Souza, Moan Jéfter Fernandes Costa, Boniek Castillo Dutra Borges

Dear Editor and Reviewer,

Thank you very much for your thorough evaluation of our manuscript. We sincerely appreciate the valuable comments and suggestions provided. As recommended, the manuscript has been carefully revised by a native English speaker to improve clarity and overall quality.

We are grateful for your time and effort in helping us enhance our work.

Kind regards,